Author's response to reviews

**Title:** Elevated expression of periostin in diabetic cardiomyopathy and the effect of valsartan

**Authors:**

jun guan (18300271751@139.com)  
mingqing xing (375816589@qq.com)  
yue shi (1156162434@qq.com)  
xueying tan (13012413468@126.com)  
chgangqiang jiang (wfjm8999@sohu.com)  
wenqi liu (laoshuaigi@163.com)  
hongyan dai (daihy9@163.com)

**Version:** 3  
**Date:** 2 March 2015

**Author's response to reviews:**

1. The abstract of the manuscript be structured into separate sections:
   - Background
   - Methods
   - Results
   - Conclusions.
2. Research Article be structured into the following sections:
   - Background
   - Methods
   - Results
   - Discussion (Results and Discussion be combined)
   - Conclusions
   - List of abbreviations used.
3. We update our ethics statement to include the name of the ethics committee that approved our study.
4. We make our manuscript include line and page numbers.
5. Authors’ Contributions section to be placed after the Competing Interests section now.
6. We provide a "Conclusions" section after the Results and Discussion section.